UC Merced UC Merced Previously Published Works

Title

Acute Experiences of Negative Interpersonal Interactions: Examining the Dynamics of Negative Mood and Ambulatory Blood Pressure Responses Among Black and Hispanic Urban Adults.

Permalink https://escholarship.org/uc/item/9zc1715k

Journal Annals of Behavioral Medicine, 57(8)

Authors

Zawadzki, Matthew Graham-Engeland, Jennifer Robles, Patrick <u>et al.</u>

Publication Date 2023-07-19

DOI

10.1093/abm/kaad033

Peer reviewed

Acute Experiences of Negative Interpersonal Interactions: Examining the Dynamics of Negative Mood and Ambulatory Blood Pressure Responses Among Black and Hispanic Urban Adults

Matthew J. Zawadzki, PhD^{1,1} · Jennifer E. Graham-Engeland, PhD² · Patrick L. Robles, MPH³ · Maryam Hussain, PhD^{1,4} · Emily V. Fair, MA² · Jonathan N. Tobin, PhD⁵ · Andrea Cassells, MPH⁵ · Elizabeth Brondolo, PhD^{6,1}

¹Department of Psychological Sciences, University of California, Merced, CA, USA ²Department of Biobehavioral Health, The Pennsylvania State University, State College, PA, USA ³Department of Sociomedical Sciences, Columbia University, New York, NY, USA ⁴Clinical Research and Development, Lucid Lane, Inc., Los Altos, CA, USA ⁵Clinical Directors Network, New York, NY, USA ⁶Department of Psychology, St. John's University, Queens, NY, USA

Matthew J. Zawadzki

mzawadzki@ucmerced.edu

Portions of this work were presented at the American Psychosomatic Society virtual conference in December 2020.

Authors' Contributions Matthew Jason Zawadzki, PhD (Conceptualization: Lead; Formal analysis: Lead; Visualization: Lead; Writing – original draft: Lead; Writing – review & editing: Lead), Jennifer E. Graham-Engeland, PhD (Conceptualization: Supporting; Writing – original draft: Supporting; Writing – review & editing: Supporting), Patrick L. Robles, MPH (Writing – original draft: Supporting; Writing – review & editing: Supporting; Writing – original draft: Supporting; Writing – review & editing: Supporting; Writing – original draft: Supporting; Writing – review & editing: Supporting; Writing – original draft: Supporting; Writing – review & editing: Supporting), Maryam Hussain, PhD (Conceptualization: Supporting; Writing – original draft: Supporting; Writing – review & editing: Supporting), Emily V. Fair, MA (Writing – original draft: Supporting), Jonathan N. Tobin, PhD (Methodology: Supporting; Project administration: Supporting), Andrea Cassells, MPH (Methodology: Supporting; Project administration: Supporting), and Elizabeth Brondolo, PhD (Conceptualization: Lead; Data curation: Lead; Funding acquisition: Lead; Methodology: Lead; Project administration: Lead; Writing – original draft: Supporting; Writing – review & editing: Supporting)

Transparency Statements In terms of transparency statements, neither this study nor the analysis plan were formally pre-registered. De-identified data from this study are not available in an a public archive. De-identified data from this study will be made available (as allowable according to institutional IRB standards) by emailing the corresponding author. Analytic code used to conduct the analyses presented in this study, as well as study materials related to the ecological momentary assessment questions, are available in a public archive: https://osf.io/d4xzk/?view_only=f3c196eff1034db98b30f9a8af98788d.

Abstract

Background Negative interpersonal interactions are associated with acute increases in ambulatory blood pressure (ABP). Yet, the mechanisms underlying this relationship are unclear.

Purpose This study tested whether negative interpersonal interactions predict higher ABP both in the moment and during subsequent observations, and whether increases in negative mood mediate these relations. These associations were tested among Black and Hispanic urban adults who may be at higher risk for negative interpersonal interactions as a function of discrimination. Race/ethnicity and lifetime discrimination were tested as moderators.

Methods Using a 24-hr ecological momentary assessment (EMA) design, 565 Black and Hispanic participants (aged 23–65, M = 39.06, SD = 9.35; 51.68% men) had their ABP assessed every 20 min during daytime accompanied by an assessment of negative interpersonal interactions and mood. This produced 12,171 paired assessments of ABP and self-reports of participants' interpersonal interactions, including how much the interaction made them feel left out, harassed, and treated unfairly, as well as how angry, nervous, and sad they felt.

Results Multilevel models revealed that more intense negative interpersonal interactions predicted higher momentary ABP. Mediation analyses revealed that increased negative mood explained the relationship between negative interpersonal interactions and ABP in concurrent and lagged analyses. Discrimination was associated with more negative interpersonal interactions, but neither race/ethnicity nor lifetime discrimination moderated findings.

Conclusions Results provide a clearer understanding of the psychobiological mechanisms through which interpersonal interactions influence cardiovascular health and may contribute to health disparities. Implications include the potential for just-in-time interventions to provide mood restoring resources after negative interactions.

© Society of Behavioral Medicine 2023. All rights reserved. For permissions, please e-mail: journals.permissions@oup.com.

Lay summary

Being mistreated by others has been shown to have negative impacts on cardiovascular health, including higher blood pressure (BP) levels. Yet, it is not clear why this mistreatment leads to increased and sustained influences on BP. In this paper, among a sample of Black and Hispanic urban adults, we studied whether changes in negative mood after being treated unfairly, excluded, or harassed explained the reason for higher BP levels. Participants completed reports of how they were treated in recent social interactions, and their levels of negative mood they were feeling at the current moment, every 20 min for 1 day. A BP measurement also occurred at each measurement. We found that negative mood was higher when a person reported being treated unfairly, excluded, and/or harassed, and that the negative mood that followed these negative interpersonal interactions accounted for increases in BP. These results have implications for how mistreatment can lead to chronic illness over time, and provides the potential for providing resources to restore mood and improve BP after mistreatment.

Keywords Ambulatory blood pressure · Negative interpersonal interaction · Social interaction · Ecological momentary assessment · Negative mood · Stress

Negative interpersonal interactions involve mistreatment in which one person withdraws positive actions and/or initiates and maintains negative actions towards another [1]. These negative interpersonal interactions can take different forms, including acts of exclusion, unfair treatment, or harassment [2]. Cross-sectional research suggests that people who report more chronic exposure to interpersonal mistreatment have worse cardiovascular profiles, including higher blood pressure (BP) [3, 4]. Laboratory studies demonstrate acute experiences of mistreatment causally relate to increases in negative mood and BP [5-7]. Yet, there are gaps in knowledge about the relationship between mistreatment and changes in BP, including limited understanding of the psychobiological correlates and consequences of daily experiences of maltreatment in everyday life. Further, despite theoretical and empirical linkages suggesting that negative mood occurs after negative interpersonal interactions and can predict BP levels, it is unclear whether changes in mood during and/or soon after negative interpersonal interactions mediate the association between such interactions and ambulatory blood pressure (ABP). Understanding the potential mediating role of mood states has clinical utility as mood states are potentially modifiable and could be a target for intervention. Thus, this paper examines the extent to which momentary negative interpersonal interactions in daily life are associated with momentary increases in ABP, and whether increases in momentary negative mood mediate such associations. We examine these associations in a sample of Black and Hispanic urban adults, people who may face higher rates of negative interpersonal interactions as a function of racial and ethnic discrimination [8, 9]. Further, Black adults are also at higher risk for hypertension and other forms of cardiovascular disease, increasing the public health significance of any such observed associations [10].

Negative Interpersonal Interactions in Everyday Life

In prior research, researchers found that higher average levels of daily negative interpersonal interactions were associated with higher average levels of daily ABP [11]. Although a strength of this past work was its use of a daily diary design, data were aggregated and analyzed at the between-person level. Such between-person research is critical to demonstrate who is at risk for future cardiovascular disease. However, between-person analyses do not tell the researcher why or when this association exists [12].

To advance a mechanistic understanding of these associations, within-person methodological approaches, such as ecological momentary assessment (EMA), are needed to determine how ABP changes when a given individual does and does not experience a negative interpersonal interaction. There is limited existing research using within-person approaches to examine the relations of negative interactions to cardiovascular reactivity and recovery in everyday life. For example, both Black and White adolescents who reported unfair treatment (e.g., being treated with less respect) throughout the day had elevated ABP in those moments when the unfair treatment occurred [3]. Other work indicates that perceived discrimination among Black and White adults predicts momentary increases in daytime ABP in response to interpersonal stress [13], suggesting that chronic exposure to unfair treatment sensitizes people to future mistreatment. With both studies, however, the trajectory of ABP responses following unfair treatment is unclear.

Is Negative Mood a Mediator?

Data from a variety of studies suggest that intrapersonal mechanisms, including negative mood, may mediate a connection between negative interpersonal interactions and elevated ABP. First, negative interpersonal interactions predict mood states. A meta-analysis found that social exclusion was followed by a consistent shift to more negative mood states [14]. Additionally, an EMA study of working adults revealed social interactions that made participants feel excluded or treated unfairly (versus other forms of interactions) were followed by reports of increased sadness [15]. Second, higher levels of negative mood are positively associated with higher levels of ABP in everyday life. For example, in a study of community adults, momentary mood that was characterized as having a more negative valence was associated with greater systolic ABP than mood in moments with less negative valence [16]. Other work has shown that more negative emotional states like anxiety [17], or a composite of negative affect [18], are associated with higher ABP levels than less intense states. However, no work to our knowledge has examined whether negative mood mediates associations between negative interpersonal interactions and ABP.

When examining mediation, it is important to consider the potential that relationships unfold over time. Negative interpersonal interactions might relate to mood when the interaction first occurs, but also influence mood for a sustained period following the interaction. In turn, both acute and sustained mood states may be linked to ABP. Theoretical explanations for the relations between negative interpersonal interactions and disease often suggest the possibility that negative interpersonal interactions evoke prolonged changes in mood and/or BP. Yet, this temporal approach to the study of the effects of negative interpersonal interactions has largely been ignored in empirical studies [19]. The use of EMA paradigms can permit us to investigate not only if negative interactions have concurrent relationships with mood and ABP, but also if these interactions result in more sustained changes in mood that are associated with elevations in ABP. These data can potentially provide novel evidence for direct and indirect pathways by which negative interpersonal interactions can lead to other downstream effects on health.

Race/Ethnicity and Discrimination as Potential Moderators

In the present research, all participants identified as either Black or Hispanic and reported on lifetime discrimination at the intake session. Past work has found that those with higher levels of lifetime discrimination reported higher levels of diastolic ABP [20, 21]. Expanding beyond this initial finding, it is possible that those who have experienced more lifetime discrimination may be sensitized to negative interpersonal interactions resulting in greater increases in BP responses to interpersonal maltreatment [22, 23]. Over time, chronically elevated BP may be, in part, a function of repeated occurrences of acute BP reactions and delayed recovery.

The Present Study

The present study utilized an EMA design in which participants reported on negative interpersonal interactions and mood every 20 min during daytime hours over a 24-hr period while ABP was concurrently assessed. The use of EMA to repeatedly assess negative social interactions and mood has the benefit of reducing recall bias and responses influenced by social desirability [24]. This is particularly important with negative interpersonal interactions because an individual's perceptions of the social interactions, which are often complex social stressors, could change over time as they cope with these events or as the situation evolves [25]. We focused on ABP as it is a unique and often superior predictor of cardiovascular disease and mortality in comparison to lab-based and clinic BP [26]. Moreover, moment-to-moment elevations in ABP are also predictive of cardiovascular disease over time [27]. We implemented 20-min time periods based on field and laboratory research demonstrating that mood [28] and BP [29] effects can last for at least this time period. Moreover, although ABP can be quick to recover in response to negative stimuli [30], the reactivity of ABP to negative stimuli makes it a strong candidate to study in relation to negative mood [16]. The proposed mediational models thus have the potential to show both the immediate and prolonged effects of negative interpersonal interactions, which may act through negative mood, to put extended burden on the cardiovascular system. As such, Research Question (RQ) 1 tested the hypothesis that there would be direct effects of negative interpersonal interactions on momentary levels of ABP. RQ2 tested the hypothesis that negative mood would mediate the relationship between negative interpersonal interactions and ABP, both in concurrent assessment models and in lagged analyses testing a cascading effect over time. RQ3 tested whether greater lifetime discrimination would be associated with a stronger connection between negative interpersonal interactions and ABP.

Methods

Participants

Participants were recruited for a study of racism, coping, and BP [31]. An initial sample of 670 participants were recruited, of whom 644 had at least some data for ABP and EMA.

Additionally, participants needed at least six ABP observations and EMAs to ensure stability of the measures and to provide a reliable person-level estimate [32]. The final sample (n = 565) consisted of participants who identified as either Black (302; 53.45%) or Hispanic (263; 46.55%) and were all U.S.-born English-speaking adults between the ages of 23 and 65 (M = 39.06, SD = 9.35); about half were men (292, 51.68%). Participants average annual income was \$20,114 (SD = \$29,860). The sample had the following levels of education: only grade school (18; 3.19%), some high school education (149; 26.37%), obtained a high school degree (142, 25.13%), some college education (119; 21.06%), technical school training (29; 5.13%), obtained a college degree (70; 12.39%), or at least some graduate school training (38; 6.73%). Finally, participants had an average body mass index (BMI) of 28.05 (SD = 5.35).

Recruitment took place in the New York City metropolitan area. Participants were recruited by Clinical Directors Network (CDN), a practice-based research network. CDN enabled us to collaborate with local community health centers and to recruit participants through their sites (e.g., visits to the waiting rooms and posting flyers). Further, a portable lab was established on an urban campus of the university of one of the authors, and flyers were posted in neighboring venues (e.g., stores, coffee shops). Participants could refer others to join the study.

Inclusion criteria included identifying as Americanborn Black/African American or American-born Hispanic/ Latino(a), being between the ages of 25–65, and being able to read and write at an 8th grade level in English or Spanish. Exclusion criteria included taking medication that affected BP, having a major medical condition that could affect ABP, or having an arm circumference greater than 44 cm (which would have precluded the use of available equipment).

Procedure and Materials

Overview and training

All study materials were approved by the Institutional Review Boards of St. John's University, CDN, Jamaica Hospital Medical Center, and the City University of New York. Participants provided written informed consent prior to participating in any study activities.

At a baseline session, participants completed a brief survey to assess demographics and had height and weight measurements taken. About two weeks after this visit, participants were fitted with a BP cuff and then had eight initial sitting and standing readings taken to train participants on the ABP measurement process and to establish baseline BP. As part of training, participants were instructed on how to terminate or initiate an ABP reading. They were also trained on EMA completion. ABP readings were then taken for the ensuing 24 hrs, every 20 min, with EMA completion occurring with each ABP reading during daytime hours (based on self-reported times when participants reported they would likely be going to sleep). They returned the equipment the next day, after which they were paid \$165 and debriefed. Additional measurements not relevant to the present study were collected at all visits. All participants were English-speaking and completed study materials in English (but had the option to complete them in Spanish). All EMA and ABP data were collected on a weekday. Data were collected between October 2003 and March 2007.

Baseline assessment

At baseline, participants provided basic demographic information, including age in years, sex (0 = woman, 1 = man), identification as Latina/o/x or Hispanic, and race identity. The ethnicity and race variables were recoded into a single category (0 = Hispanic, 1 = Black), given that only participants identifying as exclusively Black or Hispanic were included in the study. Lifetime discrimination was measured with the 34-item Perceived Ethnic Discrimination Questionnaire—Community Version [33]. This scale measures lifetime experiences of racial and ethnic discrimination within an interpersonal context. A sample item asks, "Because of your race or ethnicity, how often has someone said something disrespectful, either to your face or behind your back?." Participants responded on a scale from 1 (never) to 5 (very often). Items were averaged together such that higher numbers indicated more lifetime experiences of perceived discrimination (Cronbach's $\alpha = .95$). Finally, height and weight were measured to calculate BMI.

Ambulatory blood pressure measurement

Ambulatory blood pressure was collected via the Suntech Accutracker II (Suntech Medical Instruments, Raleigh, NC), which has shown to be a reliable and valid device [34]. To ensure the validity of readings, all data from a particular measurement were deleted if any of the following conditions were met [31, 35]: (a) the difference between systolic ABP and diastolic ABP readings was either less than 20 mm Hg or greater than 90 mm Hg, (b) a reading was accompanied by an error code from the device, and/or (c) if systolic ABP was less than 85 mm Hg or greater than 196 mm Hg, if diastolic ABP was less than 41 mm Hg or greater than 130 mm Hg, and/or if heart rate was less than 46, or greater than 130, beats per minute. There was an average of 30.89 (SD = 9.16, range = 6-58) readings per participant.

Ecological momentary assessment

After each daytime ABP measure, participants were instructed to complete the EMA. EMAs were administered using a personal device assistant (SONY CLIE) with the Quest Admin Program [36]. Assessments were date- and time-stamped. Picture icons accompanied the written questions to facilitate interpretation and response.

For negative interpersonal interactions, participants reported if they were talking with anyone when the timer went off (0 = no, 1 = yes). When participants indicated yes, they then responded on a 0-100 scale to three items assessing the degree of their perceptions of the interactions: "Did you feel [left out/ignored, harassed, treated unfairly]." These three items were correlated at moderate levels at the within-person momentary level (.53 to .59). As such, to examine an overall effect of negative interpersonal interactions, we created an average at each moment, with higher numbers indicating the interaction was perceived as more intensely negative. This scale demonstrated moderate reliability to detect within-subject differences in change over time, $R_c = .53$ [37]. In the same moments, mood was measured on a 0-100 scale asking, "When the timer went off did you feel [angry, nervous, sad]." For each assessment, the negative mood items were averaged together, with higher numbers indicating more negative mood; this scale demonstrated moderate reliability to detect change over time, $R_{c} = .52.$

To control for potential momentary influences on ABP, participants reported on their current posture from the following options: reclining (0), sitting (1), standing (2), walking (3), or running (4). Additionally, participants reported if they had eaten, smoked, drunk caffeine, and/or consumed alcohol since the last observation (i.e., 20 min earlier), with each item coded as no (0) or yes (1). There was an average of 21.65 (SD = 9.13, range = 6–42) EMA readings per participant. In the data processing stage, EMAs were merged with the accompanying ABP readings if the assessment occurred within 5 min of the reading, resulting in 12,171 paired ABP and EMA moments.

Analytic Plan

To test RQ1, we conducted multilevel models to account for the nested nature of observations within people. Models examined whether negative interpersonal interactions predicted systolic and/or diastolic ABP, testing both betweenperson and within-person effects. At the between-person level, negative interpersonal interactions were entered as an averaged value across all moments to test whether those who experienced more intense negative interpersonal interactions were also those with worse mood/ABP. At the within-person level, negative interpersonal interactions were entered as a person-mean centered value to test whether experiencing more intense negative interpersonal interactions in a moment than was typical for that person resulted in higher levels of negative mood/ABP in that moment compared to other moments with less intense negative interpersonal interactions for that person.

All models controlled for the person-level factors of age, sex, race/ethnicity, and BMI. At the momentary level, models controlled for time of day, the posture of the participant at time of measurement, and whether the participant had eaten, smoked, consumed caffeine, and/or drunk alcohol since the last measurement. A random intercept was included to allow the likelihood that participants had differing initial levels of mood and/or ABP. A random slope was specified for the within-person level of negative interpersonal interactions to allow for the possibility that the size of the relationship between negative interpersonal interactions and outcomes differed across participants. An autoregressive covariance structure was included to control for the possibility that observations closer in time to each other were more strongly related to those further apart. Finally, to estimate effect size, pseudo R^2 was calculated by estimating a predicted value for each moment that was then correlated with the actual value [38].

To test RQ2, we adopted two recommended approaches given the lack of consensus for how to perform mediation models for multilevel data. First, in line with a more traditional approach [39, 40], we conducted a series of models that tested (a) whether person-mean centered negative interpersonal interactions predicted negative mood, and (b) whether person-mean centered negative mood predicted systolic ABP and/or diastolic ABP. If significant relationships were observed, (c) we then examined a model in which both person-mean centered negative interpersonal interactions and negative mood were tested in the same model, to assess if negative interpersonal interactions had a direct relationship with ABP. All models used the same control variables as described in RQ1, including both a random intercept and random slope for the predictor of interest, and an autoregressive covariance structure. Second, we followed procedures to detect indirect effects in which the withinperson predictor and mediator were simultaneously modeled as predictors of the within-person outcome—a 1–1–1 mediation model [41]. With this approach, the data were restructured to split the negative interpersonal interactions and mood scores from the same observation to separate rows, thus creating a new data level. A multilevel model was then run with the negative interpersonal interactions and mood predicting ABP, along with a series of dummy coded terms and interaction effects that accounted for the new nested data structure. A macro was then utilized to produce indirect and total effects [42].

To test these processes over time, we ran a similar set of models but now tested negative interpersonal interactions, mood, and ABP at different time points. Specifically, we tested whether ABP at time t was predicted by mood at time t-1 that was predicted by negative interpersonal interactions at time t-2. For these lagged models, we chose not to include lagged ABP in our models in line with perspectives that this variable is very likely correlated with the random components of the multilevel model and thus would violate model assumptions [43, 44]. We continued to model a random intercept and random slope for the predictor of interest, and included the same controls as the models in RQ1.

To test RQ3, we ran two sets of models extending those tested in RQ1. First, to test race/ethnicity as a moderator, we added an interaction term of negative interpersonal interactions at the within-person level with race/ethnicity. Second, to test lifetime discrimination as a moderator, we added both the measure of lifetime discrimination (grand mean centered to aid in interpretation of any observed interaction effects), and an interaction term of negative interpersonal interactions at the within-person level and lifetime discrimination.

Results

Descriptive Statistics

Participants reported talking with someone in about half of all observations (n = 5,680, 46.44%). During these moments, participants reported low average intensity of negative interpersonal interactions but with large variation in these responses (M = 4.68, SD = 13.10). Interactions were perceived as at least minimally negative (i.e., a non-zero report on exclusion, harassment, or unfair treatment) during 25.26% (n = 1,435) of observations with a social interaction. As is typically observed, participants reported relatively low levels of negative mood (M = 6.83, SD = 14.63). Finally, on average participants had pre-hypertensive levels of systolic ABP on average (M = 133.09, SD = 20.91), but normotensive levels of diastolic ABP (M = 79.50, SD = 14.22).

RQ1: Negative Interpersonal Interactions and ABP

RQ1 examined the overall effect of negative interpersonal interactions on ABP. At the between-person level, experiencing more negative interpersonal interactions on average was not significantly related to systolic (t = 0.07, p = .94) nor diastolic ABP (t = 0.45, p = .65) (see Table 1). In contrast, at the within-person level, the intensity of negative interpersonal interactions in the moment was positively associated with the level of systolic ABP (t = 2.02, p = .044) and diastolic ABP (t = 2.31, p = .022).

RQ2: Negative Mood as a Mediator

RO2 examined whether negative mood functioned as a mediator between negative interpersonal interactions and ABP. All models controlled for the same variables as in RQ1, but only the main pathways of interest are presented in Figs. 1 and 2. Reports of the indirect and total effect are provided in Table 2. For the concurrent assessments, as shown in Fig. 1a and b, more intense levels of negative interpersonal interactions were associated with higher levels of momentary negative mood (t = 10.54, p < .001), and higher levels of momentary negative mood were associated with higher levels of systolic (t = 2.36, p = .019) and diastolic ABP (t = 3.67, p < .001). When both negative interpersonal interactions and negative mood were included in the same model, negative interpersonal interactions were no longer significantly associated with systolic (t = 1.56, p = .12) nor diastolic ABP (t = 1.70, p = .09). A test of indirect effects, displayed in Table 2, indicated negative interpersonal interactions had a significant indirect effect on diastolic ABP through negative mood (t = 3.01, p = .003), but this indirect effect was not significant for systolic ABP (t = 1.01, p = .31). This pattern of results suggests that in concurrent analyses, negative mood mediates the relationship of negative interpersonal interactions on diastolic ABP.

For the lagged assessments, as shown in Fig. 2a and b, greater negative interpersonal interactions at time t-2 predicted increased negative mood at time t-1 (t = 6.48, p < .001). In turn, higher levels of negative mood at time t-1 predicted higher time t systolic (t = 2.71, p = .007) and time t diastolic ABP (t = 2.82, p = .005). Furthermore, as reported in Table 2, the indirect effect of negative mood was significant in analyses of systolic ABP (t = 2.31, p = .021), but not diastolic ABP (t = 0.67, p = .50). This pattern of results suggest that negative mood mediates the relationship of negative interpersonal interactions on systolic ABP over time.

RQ3: Race/Ethnicity and Lifetime Discrimination as Moderators

We then explored whether lifetime discrimination predicted the level of negative interpersonal interactions at each moment. We used the models as described in RQ1, except with grand mean centered lifetime discrimination predicting levels of negative interpersonal interactions. Those who reported higher levels of lifetime discrimination also had higher levels of negative interpersonal interactions in their daily life, b =2.06, SE = 0.64, t = 3.23, CI = 0.81–3.32, p = .001.

RQ3 tested whether race/ethnicity or lifetime discrimination moderated associations tested in RQ1. All models were the same as reported earlier except an interaction term of negative interpersonal interactions by race/ethnicity (RQ3a) or lifetime discrimination and an interaction between negative interpersonal interactions by lifetime discrimination (RQ3b) was added. For race/ethnicity, the interaction term was not significant for neither systolic ABP, b = -0.02, SE = 0.06, t = -0.27, CI = -0.14 to 0.11, p = .79, nor diastolic ABP, b = -0.04, SE = 0.05, t = -0.79, CI = -0.14 to 0.06, p = -0.14.43, suggesting similar effects for Black and Hispanic participants in the study. Likewise, for lifetime discrimination, the interaction term was not significant for neither systolic ABP, b = 0.004, SE = 0.04, t = 0.09, CI = -0.08 to 0.09, p = .92, nor diastolic ABP, b = 0.02, SE = 0.03, t = 0.54, CI = -0.05to 0.09, p = .59, suggesting similar effects for those reporting higher versus lower levels of lifetime discrimination. Given Table 1 Effect of Negative Interpersonal Interactions (NIIs) on Ambulatory Blood Pressure (ABP)

	Systolic ABP			Diastolic ABP			
	b	SE	95% CI	b	SE	95% CI	
Random effects							
Intercept	152.98	14.46	-	77.72	5.94	-	
Intercept * NII	-0.14	0.44	-	-0.09	0.20	-	
NII	0.03	0.02	-	0.02	0.01	-	
Variance	112.74	8.51	_	105.13	2.28	_	
Autoregression	0.68	0.05	-	0.15	0.02	-	
Residual	110.31	8.31	_	1.02	0.00	_	
Fixed effects							
Intercept	95.66	4.36	87.10-104.23	53.70	2.92	47.97-59.44	
Age	0.12	0.07	-0.01 to 0.26	0.19	0.05	0.10-0.28	
Male	9.13	1.26	6.65-11.61	4.20	0.85	2.54-5.86	
Black Race	3.43	1.27	0.93-5.93	3.18	0.85	1.51-4.85	
BMI	0.68	0.12	0.45-0.91	0.38	0.08	0.22-0.53	
Time	0.003	0.001	0.001-0.01	0.001	0.001	-0.0004 to 0.002	
Posture	2.22	0.23	1.76-2.68	1.91	0.17	1.57-2.26	
Eating	1.17	0.49	0.21-2.14	0.76	0.37	0.04-1.49	
Smoking	1.44	0.59	0.28-2.60	1.67	0.44	0.80-2.53	
Caffeine	0.36	0.65	-0.92 to 1.63	-0.40	0.48	-1.36 to 0.54	
Alcohol	1.65	1.14	-0.58 to 3.88	2.69	0.82	1.09-4.29	
NII (Between-Person)	0.004	0.06	-0.11 to 0.12	0.02	0.04	-0.05 to 0.10	
NII (Within-Person)	0.06	0.03	0.002-0.12	0.05	0.02	0.01-0.10	
Model statistics							
Pseudo R ²	.104			.100			

Coefficients in bold significant at p < .05.



Fig. 1. Negative mood as a mediator of negative interpersonal interactions and ambulatory blood pressure (ABP).



Fig. 2. Time T-1 negative mood as a mediator of time T-2 negative interpersonal interactions and time T ABP.

Table 2 Indirect and Total Effects of Negative Interpersonal Interactions of Systolic and Diastolic ABP through Negative Mood

	Indirect effect			Total effect		
	b	SE	CI	b	SE	CI
Concurrent models						
Negative Interpersonal Interaction \rightarrow Negative Mood \rightarrow Systolic ABP	0.02	0.02	-0.02 to 0.15	0.08	0.04	0.01-0.15
Negative Interpersonal Interaction \rightarrow Negative Mood \rightarrow Diastolic ABP	0.04	0.01	0.01-0.06	0.06	0.03	0.01-0.11
Lagged models						
Negative Interpersonal Interaction _{<i>t</i>-2} \rightarrow Negative Mood _{<i>t</i>-1} \rightarrow Systolic ABP _{<i>t</i>}	0.02	0.01	0.004-0.04	0.03	0.04	-0.04 to 0.10
Negative Interpersonal Interaction _{$t-2 \rightarrow Negative Mood_{t-1} \rightarrow Diastolic ABP_t$}	0.01	0.01	-0.01 to 0.03	0.004	0.03	-0.04 to 0.05

Coefficients in bold significant at p < .05.

the lack of moderation for RQ1, we did not consider additional moderation effects for RQ2.

Discussion

Although prior research has demonstrated that certain types of negative interpersonal interactions can affect ABP and negative mood, we extended this work to examine both ABP reactivity and recovery from negative interpersonal interactions. This research is particularly novel in examining whether changes in mood helped explain connections between negative interpersonal interactions and ABP. We examined these associations in a sample of Black and Hispanic participants, who historically are at high risk for hypertension [45] and increased risk of experiencing negative interpersonal interactions as a function of racial and ethnic discrimination [8, 9], which we also observed in the present study.

We first tested direct relationships between negative interpersonal interactions and ABP. As expected, higher levels of momentary negative interpersonal interactions were associated with higher ABP. These findings support and extend prior work demonstrating the activating effects of negative interpersonal interactions [3, 15, 46]. These findings also help demonstrate how common these experiences are, with negative interpersonal interactions occurring around 12% of all moments, or a bit more than three experiences per each 24-hr assessment period. The relatively high frequency of negative interpersonal interactions in the present research may be in part a function of our use of EMA to capture nuanced and in-the-moment episodes in daily life rather than relying on retrospective reports that may focus on more memorable, recent, or unresolved events, for example, [47]. Other research using EMA to examine experiences of sexist experiences also reported a relatively high frequency of discrimination [48]. The frequency of negative interpersonal interactions in daily life is likely important, as consequences may accrue to predict risk. For example, the frequency of negative interactions has been found to mediate the effects of perceived discrimination on health outcomes [49].

A second goal of the present research was to test the hypothesis that emotional responses help explain how negative interpersonal interactions are related to ABP, both at a given moment in a day and later in the same day. Although past work suggests that negative interpersonal interactions can function as a stressor [13], causing emotional and cardiovascular reactivity [14], to our knowledge no past research has examined changes in negative mood as a mediator of negative interpersonal interactions on ABP in daily life. In our concurrent mediation analyses, negative mood mediated the relationship between negative interpersonal interactions and diastolic (but not systolic) ABP. In the lagged mediation analyses, although there were no direct associations between negative interpersonal interactions and ABP assessed later in the day (i.e., up to 40 min later), there was an indirect relationship through negative mood assessed up to 20 min after the negative interpersonal interaction on systolic (but not diastolic) BP. This pattern of results indicates that the effects of negative interpersonal interactions on BP are sustained due to the lack of affective recovery following the interaction, and suggests complex patterns of emotion regulation that evolve over time, influencing the relations of social interactions, mood, and ABP. As such, there may be other possible mechanisms that buffer BP reactivity and/or aid in BP recovery, including positive mood, that have demonstrated a protective role for cardiovascular health [50, 51].

Lastly, we tested whether participants' race/ethnicity or reported lifetime discrimination moderated associations but did not find any significant moderation effects. Given prior work suggesting that past discrimination may sensitize people to experiences of mistreatment and predict greater reactivity [22, 23], we hypothesized that individuals who reported greater lifetime discrimination in the current study might show greater reactivity to negative interpersonal interactions. In prior research, researchers were able to connect attributions for the negative interpersonal mistreatment to one's race or ethnicity. In everyday life, it is possible that higher lifetime exposure to discrimination makes one more likely to anticipate or expect to be treated in a discriminatory manner [52], but not necessarily to have greater physiological reactivity to all negative interactions. That said, we found that higher levels of lifetime discrimination predicted higher levels of negative interpersonal mistreatment. This suggests that, contrary to some extant work [53], the experience of discrimination in the past does not desensitize an individual to the ongoing experience of discrimination (and perhaps may even sensitize one to noticing negative experiences [8], or that ongoing discriminatory experiences are occurring). Future research is warranted to examine nuances in the extent to which lifetime discrimination influences affective and cardiovascular responses to negative interpersonal interactions, including testing the potential for an indirect effect of lifetime discrimination through everyday interactions.

Overall, the present findings likely have clinical implications given the strong link between ABP and health risk [26, 27]. Our analyses highlight the need to better understand the impact of negative interpersonal interactions on both mood and ABP and to examine in-the-moment coping mechanisms. To this end, an important next step would be to uncover additional modifiable psychological, cognitive, or behavioral mechanisms that lead to sustained negative effects of interpersonal maltreatment. For example, experiencing mistreatment may evoke negative mood and trigger ruminative or vigilant states that maintain negative mood and make a future negative interpersonal interaction more likely [54]. Another complementary possibility is that individuals experiencing negative interpersonal interactions engage in social withdrawal, preventing additional conflict but limiting the ability to acquire skills in emotion regulation. Moreover, coping strategies that engage more social support may improve cardiovascular outcomes [55]. Social and behavioral strategies at a dyadic or societal level are needed to prevent the occurrence of, or minimize the effects of, negative interpersonal interactions that have downstream effects on cardiovascular risk.

Limitations and Future Directions

As is common for this type of EMA research, ABP in this study was assessed frequently (every 20 min) with a selfreport at each assessment. This approach allowed for a tight window between any reported negative interpersonal interaction and the subsequent assessment of mood and ABP. However, this frequency of assessment could have changed the way people interacted with others. Due to the frequency of assessments, the EMA measures were brief to reduce participant burden, including the three-item negative mood scale, which may have been less reliable as a measure than a scale with additional items [56]. Our models had relatively small effect sizes in predicting ABP, although negative interpersonal interactions had independent effects similar to a range of known and strong influences on BP. This suggests that additional variables are needed to further consider when and for whom negative interpersonal interactions relate to cardiovascular outcomes. For example, researchers could capture the extent to which a person attributed the reason for their unfair treatment to a social identity, including their race and ethnicity. This momentary measure of attributions would allow for the comparison of whether discriminatory experiences have more negative psychobiological consequences than other types of negative interpersonal interactions. Future research could also measure and examine a wider range of contextual factors that may influence the cardiovascular consequences of negative interactions, including the social context in which the interactions occur [57], and individual differences factors, including personality, which may interact to affect responses to interpersonal interactions [58]. To this end, although we had a diverse sample of participants who identified as Black or Hispanic, they all came from a large Northeastern city in the USA. Researchers should use caution when generalizing these findings to other samples.

Conclusion

These results extend a growing literature on the deleterious effects of negative interpersonal interactions. In keeping with the Science of Behavior Change approach of identifying mechanisms of action [59], negative mood was examined as potential psychological mechanisms contributing to health outcomes. Higher levels of negative mood during these negative interpersonal interactions appeared to explain associations between interactions and ABP, both concurrently and over time. Although there is more work to be done to further uncover the nature of these relationships, these findings point to possible intervention targets, including reducing negative interactions and enhancing stress regulation.

Compliance with Ethical Standards

Conflicts of Interest No authors have conflicts of interest to declare.

Funding This work was partly supported by a grant award to Dr. Elizabeth Brondolo by the National Heart, Lung, and Blood Institute (R01 HL068590). Emily Fair was supported in part by the Alfred P. Sloan Foundation Minority Scholar program, and by NIH training grant T32 LM12415-5.

Ethical Adherence All study materials were approved by the Institutional Review Boards of St. John's University, Clinical Directors Network, Jamaica Hospital Medical Center, and the City University of New York.

References

- Andersson LM, Pearson CM. Tit for tat? The spiraling effect of incivility in the workplace. *Acad Manage Rev.* 1999; 24(3):452–471. doi:10.5465/amr.1999.2202131.
- Tepper BJ, Henle CA. A case for recognizing distinctions among constructs that capture interpersonal mistreatment in work organizations. J Organ Behav. 2011; 32(3):487–498. doi:10.1002/ job.688.
- Matthews KA, Salomon K, Kenyon K, Zhou F. Unfair treatment, discrimination, and ambulatory blood pressure in Black and White adolescents. *Health Psychol.* 2005; 24(3):258–265. doi:10.1037/0278-6133.24.3.258.
- Newton TL, Weigel RA. Cardiovascular correlates of interpersonal mistreatment in healthy adults. *J Emot Abuse*. 2007; 7(2):35–58. doi:10.1300/J135v07n02_03.
- Williamson TJ, Thomas KS, Eisenberger NI, Stanton AL. Effects of social exclusion on cardiovascular and affective reactivity to a socially evaluative stressor. *Int J Behav Med.* 2018; 25(4):410–420. doi:10.1007/s12529-018-9720-5.
- García-León A, Reyes del Paso GA, Robles H, Vila J. Relative effects of harassment, frustration, and task characteristics on cardiovascular reactivity. *Int J Psychophysiol.* 2003; 47(2):159–173. doi:10.1016/S0167-8760(02)00124-1.
- Glynn LM, Christenfeld N, Gerin W. Recreating cardiovascular responses with rumination: The effects of a delay between harassment and its recall. *Int J Psychophysiol.* 2007; 66(2):135–140. doi:10.1016/j.ijpsycho.2007.03.018.
- Broudy R, Brondolo E, Coakley V, et al. Perceived ethnic discrimination in relation to daily moods and negative social interactions. J Behav Med. 2007; 30(1):31–43. doi:10.1007/s10865-006-9081-4.
- Smith IA, Griffiths A. Microaggressions, everyday discrimination, workplace incivilities, and other subtle slights at work: A metasynthesis. *Hum Resour Dev Rev.* Published online May 14, 2022; 21(3):153448432210987. doi:10.1177/15344843221098756.
- 10. Carnethon MR, Pu J, Howard G, et al. Cardiovascular health in African Americans: A scientific statement from the American Heart Association. *Circulation*. 2017; 136(21):e393–e423. doi:10.1161/CIR.00000000000534.
- Schoenthaler AM, Schwartz J, Cassells A, Tobin JN, Brondolo E. Daily interpersonal conflict predicts masked hypertension in an urban sample. *Am J Hypertens*. 2010; 23(10):1082–1088. doi:10.1038/ajh.2010.141.
- Zawadzki MJ, Smyth JM, Sliwinski MJ, Ruiz JM, Gerin W. Revisiting the lack of association between affect and physiology: Contrasting between-person and within-person analyses. *Health Psychol.* 2017; 36(8):811–818. doi:10.1037/hea0000466.
- Smart Richman L, Pek J, Pascoe E, Bauer DJ. The effects of perceived discrimination on ambulatory blood pressure and affective responses to interpersonal stress modeled over 24 hours. *Health Psychol.* 2010; 29(4):403–411. doi:10.1037/a0019045.
- 14. Blackhart GC, Nelson BC, Knowles ML, Baumeister RF. Rejection elicits emotional reactions but neither causes immediate distress nor lowers self-esteem: A meta-analytic review of 192 studies on

social exclusion. *Personal Soc Psychol Rev.* 2009; 13(4):269–309. doi:10.1177/1088868309346065.

- Bernstein MJ, Zawadzki MJ, Juth V, Benfield JA, Smyth JM. Social interactions in daily life: Within-person associations between momentary social experiences and psychological and physical health indicators. *J Soc Pers Relatsh.* 2018; 35(3):372–394. doi:10.1177/0265407517691366.
- Zawadzki MJ, Mendiola J, Walle EA, Gerin W. Between-person and within-person approaches to the prediction of ambulatory blood pressure: The role of affective valence and intensity. *J Behav Med.* 2016; 39(5):757–766. doi:10.1007/s10865-016-9746-6.
- Edmondson D, Arndt J, Alcántara C, Chaplin W, Schwartz JE. Self-esteem and the acute effect of anxiety on ambulatory blood pressure. *Psychosom Med.* 2015; 77(7):833–841. doi:10.1097/ PSY.00000000000219.
- Smith TW, Birmingham W, Uchino BN. Evaluative threat and ambulatory blood pressure: Cardiovascular effects of social stress in daily experience. *Health Psychol.* 2012; 31(6):763–766. doi:10.1037/a0026947.
- Cole MS, Shipp AJ, Taylor SG. Viewing the interpersonal mistreatment literature through a temporal lens. Organ Psychol Rev. 2016; 6(3):273–302. doi:10.1177/2041386615607095.
- Beatty Moody DL, Waldstein SR, Tobin JN, Cassells A, Schwartz JC, Brondolo E. Lifetime racial/ethnic discrimination and ambulatory blood pressure: The moderating effect of age. *Health Psychol.* 2016; 35(4):333–342. doi:10.1037/hea0000270.
- Pascoe EA, Smart Richman L. Perceived discrimination and health: A meta-analytic review. *Psychol Bull.* 2009; 135(4):531–554. doi:10.1037/a0016059.
- Busse D, Yim IS, Campos B. Social context matters: Ethnicity, discrimination and stress reactivity. *Psychoneuroendocrinology*. 2017; 83:187–193. doi:10.1016/j.psyneuen.2017.05.025.
- Guyll M, Matthews KA, Bromberger JT. Discrimination and unfair treatment: Relationship to cardiovascular reactivity among African American and European American women. *Health Psychol.* 2001; 20(5):315–325. doi:10.1037/0278-6133.20.5.315.
- 24. Stone AA, Shiffman S. Ecological momentary assessment (EMA) in behavioral medicine. Ann Behav Med. 1994; 16(3):199–202. doi:10.1093/abm/16.3.199.
- 25. Yang F, Lu M, Huang X. Customer mistreatment and employee well-being: A daily diary study of recovery mechanisms for frontline restaurant employees in a hotel. *Int J Hosp Manag.* 2020; 91:102665. doi:10.1016/j.ijhm.2020.102665.
- 26. Kikuya M, Ohkubo T, Asayama K, et al. Ambulatory blood pressure and 10-year risk of cardiovascular and noncardiovascular mortality: The Ohasama study. *Hypertension*. 2005; 45(2):240–245. doi:10.1161/01.HYP.0000152079.04553.2c.
- Treiber FA, Kamarck T, Schneiderman N, Sheffield D, Kapuku G, Taylor T. Cardiovascular reactivity and development of preclinical and clinical disease states. *Psychosom Med.* 2003; 65(1):46–62.
- Scott SB, Sliwinski MJ, Blanchard-Fields F. Age differences in emotional responses to daily stress: The role of timing, severity, and global perceived stress. *Psychol Aging*. 2013; 28(4):1076–1087. doi:10.1037/a0034000.
- Zawadzki MJ, Gavrilova L. All the lonely people: Comparing the effects of loneliness as a social stressor to non-lonely stress on blood pressure recovery. *Int J Psychophysiol.* 2021; 167:94–101. doi:10.1016/j.ijpsycho.2021.06.016.
- Linden WLET, Earle TL, Gerin W, Christenfeld N. Physiological stress reactivity and recovery: Conceptual siblings separated at birth? J Psychosom Res. 1997; 42(2):117–135. doi:10.1016/ S0022-3999(96)00240-1.
- Brondolo E, Libby DJ, Denton E, et al. Racism and ambulatory blood pressure in a community sample. *Psychosom Med.* 2008; 70(1):49–56. doi:10.1097/PSY.0b013e31815ff3bd.
- Llabre MM, Ironson GH, Spitzer SB, Gellman MD, Weidler DJ, Schneiderman N. Blood pressure stability of normotensives and mild hypertensives in different settings. *Health Psychol.* 1988; 7(Suppl):127–137. doi:10.1037/h0090276.

- 33. Brondolo E, Kelly KP, Coakley V, et al. The Perceived Ethnic Discrimination Questionnaire: Development and preliminary validation of a community version. J Appl Soc Psychol. 2005; 35(2):335–365. doi:10.1111/j.1559-1816.2005.tb02124.x.
- White WB, Morganroth J. Usefulness of ambulatory monitoring of blood pressure in assessing antihypertensive therapy. *Am J Cardiol.* 1989; 63(1):94–98. doi:10.1016/0002-9149(89)91083-7.
- Brondolo E, Karlin W, Alexander K, Bobrow A, Schwartz J. Workday communication and ambulatory blood pressure: Implications for the reactivity hypothesis. *Psychophysiology*. 1999; 36(1):86–94. doi:10.1017/S0048577299961565.
- 36. Brondolo E, Brady N, Thompson S, et al. Perceived racism and negative affect: Analyses of trait and state measures of affect in a community sample. J Soc Clin Psychol. 2008; 27(2):150–173. doi:10.1521/jscp.2008.27.2.150.
- Bolger N, Laurenceau JP. Intensive Longitudinal Methods: An Introduction to Diary and Experience Sampling Research. New York: Guilford Press; 2013.
- Singer JD, Willett JB. Applied Longitudinal Data Analysis: Modeling Change and Event Occurrence. USA: Oxford University Press; 2003.
- Krull JL, MacKinnon DP. Multilevel modeling of individual and group level mediated effects. *Multivar Behav Res.* 2001; 36(2):249– 277. doi:10.1207/S15327906MBR3602_06.
- Zhang Z, Zyphur MJ, Preacher KJ. Testing multilevel mediation using hierarchical linear models: Problems and solutions. Organ Res Methods. 2009; 12(4):695–719. doi:10.1177/10944281 08327450.
- 41. Bauer DJ, Preacher KJ, Gil KM. Conceptualizing and testing random indirect effects and moderated mediation in multilevel models: New procedures and recommendations. *Psychol Methods*. 2006; 11(2):142–163. doi:10.1037/1082-989X.11.2.142.
- Bauer DJ. *Publications*. The University of North Carolina at Chapel Hill. https://dbauer.web.unc.edu/publications/. Accessibility verified June 16, 2022.
- 43. Allison P. Don't put lagged dependent variables in mixed models. Statistical Horizons. Published June 3, 2015. https:// statisticalhorizons.com/lagged-dependent-variables/. Accessibility verified June 16, 2022.
- 44. Spencer NH. Combining modelling strategies to analyse teaching styles data. *Qual Quant.* 2002; 36(2):113–127. doi:10.102 3/A:1014969805388.
- Lackland DT. Racial differences in hypertension: Implications for high blood pressure management. *Am J Med Sci.* 2014; 348(2):135– 138. doi:10.1097/MAJ.00000000000308.
- 46. Rook KS. Emotional health and positive versus negative social exchanges: A daily diary analysis. *Appl Dev Sci.* 2001; 5(2):86–97. doi:10.1207/S1532480XADS0502_4.

- Potter LN, Brondolo E, Smyth JM. Biopsychosocial correlates of discrimination in daily life: A review. *Stigma Health*. 2019; 4(1):38–61. doi:10.1037/sah0000120.
- 48. Swim JK, Hyers LL, Cohen LL, Ferguson MJ. Everyday sexism: Evidence for its incidence, nature, and psychological impact from three daily diary studies. J Soc Issues. 2001; 57(1):31–53. doi:10.1111/0022-4537.00200.
- 49. Park IJK, Wang L, Williams DR, Alegría M. Does anger regulation mediate the discrimination–mental health link among Mexicanorigin adolescents? A longitudinal mediation analysis using multilevel modeling. *Dev Psychol.* 2017; 53(2):340–352. doi:10.1037/ dev0000235.
- Boehm JK, Chen Y, Qureshi F, et al. Positive emotions and favorable cardiovascular health: A 20-year longitudinal study. *Prev Med.* 2020; 136:106103. doi:10.1016/j.ypmed.2020.106103.
- 51. Sin NL. The protective role of positive well-being in cardiovascular disease: Review of current evidence, mechanisms, and clinical implications. *Curr Cardiol Rep.* 2016; 18(11):106. doi:10.1007/ s11886-016-0792-z.
- Conkel-Ziebell JL, Gushue GV, Turner SL. Anticipation of racism and sexism: Factors related to setting career goals for urban youth of color. J Couns Psychol. 2019; 66(5):588–599. doi:10.1037/ cou0000357.
- Ortiz SM. "You can say I got desensitized to it": How men of color cope with everyday racism in online gaming. *Sociol Perspect*. 2019; 62(4):572–588. doi:10.1177/0731121419837588.
- 54. Birk JL, Cornelius T, Edmondson D, Schwartz JE. Duration of perseverative thinking as related to perceived stress and blood pressure: An ambulatory monitoring study. *Psychosom Med.* 2019; 81(7):603–611. doi:10.1097/PSY.000000000000727.
- 55. Saban KL, Motley D, Shawahin L, et al. Preliminary evidence for a race-based stress reduction intervention for Black women at risk for cardiovascular disease. *Complement Ther Med.* 2021; 58:102710. doi:10.1016/j.ctim.2021.102710.
- Scott SB, Sliwinski MJ, Zawadzki M, et al. A coordinated analysis of variance in affect in daily life. *Assessment.* 2020; 27(8):1683– 1698. doi:10.1177/1073191118799460.
- Beattie L, Griffin B. Accounting for within-person differences in how people respond to daily incivility at work. J Occup Organ Psychol. 2014; 87(3):625–644. doi:10.1111/joop.12067.
- Brondolo E, Rieppi R, Erickson SA, et al. Hostility, interpersonal interactions, and ambulatory blood pressure. *Psychosom Med.* 2003; 65(6):1003–1011. doi:10.1097/01.PSY.0000097329.53585.A1.
- 59. Nielsen L, Riddle M, King JW, et al; NIH Science of Behavior Change Implementation Team. The NIH Science of Behavior Change Program: Transforming the science through a focus on mechanisms of change. *Behav Res Ther.* 2018; 101:3–11. doi:10.1016/j.brat.2017.07.002.